





Mycobacterium ulcerans low infectious dose and mechanical transmission support insect bites and puncturing injuries in the spread of Buruli ulcer

Wallace, John R.; Mangas, Kirstie M.; Porter, Jessica L.; Marcsisin, Renee; Pidot, Sacha J.; Howden, Brian; Omansen, Till; Zeng, Weiguang; Axford, Jason K.; Johnson, Paul D. R.

Published in: PLoS Neglected Tropical Diseases

DOI: 10.1371/journal.pntd.0005553

IMPORTANT NOTE: You are advised to consult the publisher's version (publisher's PDF) if you wish to cite from it. Please check the document version below.

Document Version Publisher's PDF, also known as Version of record

Publication date: 2017

Link to publication in University of Groningen/UMCG research database

Citation for published version (APA): Wallace, J. R., Mangas, K. M., Porter, J. L., Marcsisin, R., Pidot, S. J., Howden, B., ... Stinear, T. P. (2017). Mycobacterium ulcerans low infectious dose and mechanical transmission support insect bites and puncturing injuries in the spread of Buruli ulcer. PLoS Neglected Tropical Diseases, 11(4). DOI: 10.1371/journal.pntd.0005553

Copyright

Other than for strictly personal use, it is not permitted to download or to forward/distribute the text or part of it without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license (like Creative Commons).

Take-down policy If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Downloaded from the University of Groningen/UMCG research database (Pure): http://www.rug.nl/research/portal. For technical reasons the number of authors shown on this cover page is limited to 10 maximum.



Citation: Wallace JR, Mangas KM, Porter JL, Marcsisin R, Pidot SJ, Howden B, et al. (2017) *Mycobacterium ulcerans* low infectious dose and mechanical transmission support insect bites and puncturing injuries in the spread of Buruli ulcer. PLoS Negl Trop Dis 11(4): e0005553. https://doi. org/10.1371/journal.pntd.0005553

Editor: Andrew S. Azman, Johns Hopkins Bloomberg School of Public Health, UNITED STATES

Received: November 23, 2016

Accepted: April 4, 2017

Published: April 14, 2017

Copyright: © 2017 Wallace et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: This research was supported by grants from the National Health and Medical Research Council of Australia (www.nhmrc.gov.au) (TPS, PDRJ: GNT1049183), the End Buruli ulcer Alliance (JRW: 6032305815) and the Millersville University Faculty grant program (JRW). The funders had no RESEARCH ARTICLE

Mycobacterium ulcerans low infectious dose and mechanical transmission support insect bites and puncturing injuries in the spread of Buruli ulcer

John R. Wallace¹*, Kirstie M. Mangas², Jessica L. Porter², Renee Marcsisin², Sacha J. Pidot², Brian Howden², Till F. Omansen^{2,3}, Weiguang Zeng², Jason K. Axford⁴, Paul D. R. Johnson⁵, Timothy P. Stinear²*

1 Department of Biology, Millersville University, Millersville, PA, United States of America, 2 Department of Microbiology and Immunology, at the Peter Doherty Institute for Infection and Immunity, University of Melbourne, Melbourne, Australia, 3 Department of Internal Medicine, University of Groningen, Groningen, RB, The Netherlands, 4 Pest and Environmental Adaptation Research Group, Bio21 Institute and School of BioSciences, University of Melbourne, Parkville, Vic, Australia, 5 Department of Infectious Diseases, Austin Health, Heidelberg, Victoria, Australia

* tstinear@unimelb.edu.au (TPS); john.wallace@millersville.edu (JRW)

Abstract

Addressing the transmission enigma of the neglected disease Buruli ulcer (BU) is a World Health Organization priority. In Australia, we have observed an association between mosquitoes harboring the causative agent, *Mycobacterium ulcerans*, and BU. Here we tested a contaminated skin model of BU transmission by dipping the tails from healthy mice in cultures of the causative agent, *Mycobacterium ulcerans*. Tails were exposed to mosquito (*Aedes notoscriptus* and *Aedes aegypti*) blood feeding or punctured with sterile needles. Two of 12 of mice with *M. ulcerans* contaminated tails exposed to feeding *A. notoscriptus* mosquitoes developed BU. There were no mice exposed to *A. aegypti* that developed BU. Eighty-eight percent of mice (21/24) subjected to contaminated tail needle puncture developed BU. Mouse tails coated only in bacteria did not develop disease. A median incubation time of 12 weeks, consistent with data from human infections, was noted. We then specifically tested the *M. ulcerans* infectious dose-50 (ID50) in this contaminated skin surface infection model with needle puncture and observed an ID50 of 2.6 colony-forming units. We have uncovered a biologically plausible mechanical transmission mode of BU via natural or anthropogenic skin punctures.

Author summary

Buruli ulcer is a neglected tropical disease caused by infection with *Mycobacterium ulcerans*. Unfortunately, how people contract this disease is not well understood. Here we show for the first time using experimental infections in mice that a very low dose of *M*. *ulcerans* delivered beneath the skin by a minor injury caused by a blood-feeding insect role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

(mosquito) or an experimental needle puncture is sufficient to cause Buruli ulcer. This research provides important laboratory evidence to advance our understanding of Buruli ulcer disease transmission.

Introduction

Among the 17 neglected tropical diseases the World Health Organization (WHO) has targeted for control and elimination, only Leprosy and Buruli ulcer (BU) have unknown modes of transmission [1]. The search to understand how humans contract BU spans more than 70 years since the causative agent, *Mycobacterium ulcerans*, was first identified [2]. There are persistent and emerging foci of BU cases across the world, in particular Africa and Australia [3]. BU is characterized by necrotizing skin lesions, caused by localized proliferation of *M. ulcerans* in subcutaneous tissue. BU is rarely fatal, but untreated infections leave patients with significant disfigurement and disability, with damaging personal and economic consequences [4, 5]. Researchers have long been struck by the characteristic epidemiology of BU, with cases occurring in highly geographically circumscribed regions (sometimes less than a few square kilometres) and risk factors for infection that include gardening, insect bites and proximity to (but not necessarily contact with) lacustrine/riverine regions [6–14]. Human-to-human spread is considered unlikely [14]. Disease transmission is thought to occur by contact with an environment contaminated with *Mycobacterium ulcerans* but exactly where the pathogen resides and why it appears so geographically restricted have yet to be determined. [15].

M. ulcerans is very slow growing (doubling time >48 hrs) and this poses a problem for source tracking efforts as it is difficult to isolate the bacteria in pure culture from complex environmental specimens [16]. M. ulcerans has only once been isolated from a non-clinical source, an aquatic water bug (Gerridae) from Benin [16]. Quantitative PCR targeting M. ulcerans-specific DNA is the most frequently used technique in surveys of environmental specimens. A comprehensive review of the many field and lab studies that have examined transmission of BU has highlighted the range of organisms from aquatic plants, snails, insects, fish, amphibia, and in Australia certain native marsupials that can serve as potential reservoirs for *M. ulcerans* [15, 17–20]. Since the first observation that biting aquatic insects can harbour *M. ulcerans* [21], studies of BU transmission have largely focused on the potential for insects to biologically vector *M. ulcerans* implying that *M. ulcerans* undergoes a propagative or reproductive mode of development in an insect [22-26]. Several case-control studies, including from both Australia and Africa have suggested insects may play a role in transmission [10, 11]. For example, in southeastern Australia, we noted Buruli lesions on exposed areas likely to attract biting insects, some patients with every brief exposure times to endemic areas [27, 28] and in 2004 we began a study that identified *M. ulcerans* DNA associated with mosquitoes captured in endemic areas [22]. However, there is no compelling experimental evidence for single-mode biological transmission of M. ulcerans via insect vectors.

Materials and methods

Ethics statement

The animal ethics committee (AEC) of the University of Melbourne approved all animal experiments under approval number AEC: 1312775.2, in accordance with the National Health and Medical Research Council Australian code for the care and use of animals for scientific purposes 8th edition (2013).

Bacterial isolates and culture conditions

M. ulcerans strain JKD8049 and bioluminescent *M. ulcerans* JKD8049 (harbouring plasmid pMV306 *hsp:lux*G13) [29, 30] were cultured in 7H9 broth or Middlebrook 7H10 agar, containing 10% oleic-albumin-dextrose-catalase growth supplement (Middlebrook, Becton Dickinson, Sparks, MD, USA) and 0.5% glycerol (v/v) at 30°C. Colony counts from bacterial cultures or tissue specimens were performed using spot plating. Five x 3µl volumes of serial 10-fold dilutions $(10^{-1} \text{ to } 10^{-5})$ of a culture or tissue preparation were spotted onto 7H10 agar plates with a 5x5 grid marked. The spots were allowed to dry, the plates loosely wrapped in plastic bags and then incubated as above for 10 weeks before counting colonies. Data analysis was performed using GraphPad Prism v 7.0a. All culture extracts were screened by LC-MS for the presence of mycolactones as previously described to ensure bacteria used in transmission experiments remained fully virulent [31].

Experimental animals

BALB/c mice were purchased from ARC (Canning Vale, Australia) and housed in individual ventilated cages. Upon arrival, animals were acclimatizing for 5 days. Food and water were given *ad libitum*.

Aedes notoscriptus and Aedes aegypti rearing

Wild caught mosquitoes were sourced from around Cairns, Queensland, Australia. *A. noto-scriptus* and *A. aegypti* colonies were reared in a Physical Containment Level 2 (PC2) laboratory environment at 26°C using previously described methods, with the addition of brown paper used as the oviposition substrate for *A. notoscriptus* [32]. Females were aged for at least one week prior to blood feeding ensure maturity, during which they were mated (seminal accessory fluid is required to stimulate searching and feeding behavior in females). The adults were provided with access to a 10% sucrose solution, which was withdrawn 24h prior to a blood feeding experiment.

Mosquito-mouse transmission experiments

Six-week old female BALB/c mice were anaesthetized and their tails coated in a thin film of M. *ulcerans* wild type strain JKD8049 by dipping the tails in a Petri dish containing 20mL of bacterial culture (concentration ~10⁶ CFU/mL) (Table 1). Tails were allowed to air-dry for 5 minutes after dipping. The tail only was then exposed to a 200mm x 200mm x 200mm cage of 20 adult female mosquitoes for a period of 15 minutes. Twenty mosquitoes were used per feeding bout to minimize stress on the mice. The number of insects biting each mouse was recorded over the exposure period by continuous observation. Tails were not wiped or disinfected postbiting. Mice were then observed weekly for up to six months for signs of tail lesions. Sterile needle stick (25G or 30G needle) and no-trauma were used as controls. An additional control consisted of tails dipped in sterile culture broth only and subjected to mosquito biting or sterile needle stick (See Table 1 for experiment design). The concentration of bacteria used was calculated by dilution plating as described above. Contingency tables and Fisher's exact test were used to test if instances of mosquito-associated or needle stick-associated disease transmission were significantly different to no trauma (GraphPad Prism v 7.0a).

Infectious dose calculations and infectious dose 50 experiment

To estimate the infectious dose, we first measured the surface area of five dissected naïve mouse-tails to obtain an average surface area (493.3 \pm 41.1 mm²). Using ten naïve mouse-tails

Trauma source	Number of mice	Mouse tail coated	Number of mice bitten	Number of mice developing BU ³	Estimated dose (CFU)
Experiment 1:					
Aedes notoscriptus	12 (4/mosquito cage ¹)	<i>M. ulcerans</i> (4.1x 10 ⁶ CFU/mL)	6	2	21
Aedes notoscriptus	4 (1/mosquito cage)	Media-only	2	0	-
Sterile needle (25G)	5	<i>M. ulcerans</i> (4.1x 10 ⁶ CFU/mL)	-	4	55
None	6	<i>M. ulcerans</i> (4.1x 10 ⁶ CFU/mL)	-	0	-
Experiment 2:			· · · · · · · · · · · · · · · · · · ·		
Aedes aegypti	5 (1/mosquito cage)	<i>M. ulcerans</i> (1.83 x 10 ⁶ CFU/mL)	5 ²	0	9
Aedes aegypti	3 (1/mosquito cage)	Media-only	2	0	-
Sterile needle (25G)	5	<i>M. ulcerans</i> (1.83 x 10 ⁶ CFU/mL)	-	4	40
None	5	<i>M. ulcerans</i> (1.83 x 10 ⁶ CFU/mL)	-	0	-
Experiment 3:					
Sterile needle (30G)	14	<i>M. ulcerans</i> (3.9 x 10 ⁶ CFU/mL)	-	13	14
None	8	<i>M. ulcerans</i> $(3.9 \times 10^6 \text{ CFU/mL})$	-	0	_

Table 1. Summary of transmission experiments.

Notes

¹There were 20 adult female mosquitoes per cage

²Multiple bites per mouse with 2 mice receiving 3 bites and 1 mouse receiving 2 bites. All other mice received only one bite.

³IS2404 qPCR results between mosquito and needle puncture experiments were not significantly different (average Ct = 20.22, range 16.52–33.5).

https://doi.org/10.1371/journal.pntd.0005553.t001

and a precision balance, we then calculated the average volume of M. ulcerans 7H9 Middlebrook culture adhering to the tail surface $(32.4 \pm 4.2 \,\mu\text{L})$, the concentration of bacteria in the cultures used, and the surface area of the tips of 25G and 30G needles used to deliver the puncture wounds (0.207 mm² and 0.056 mm² respectively). These parameters were then used to calculate the infectious dose, assuming the bacteria were evenly distributed over the tail surface (Fig 1B). A standard protocol to calculate an ID50 was followed. Four-week old female BALB/ c mice were anaesthetized and their tails coated as described above for the transmission experiments with 10-fold dilutions of bioluminescent M. ulcerans JKD8049 from 10⁶ to 10 CFU/mL in each Petri dish, using five mice per dilution, followed by a single needle stick puncture with a 25G needle. The mice were monitored using a Lumina XRMS Series III In Vitro Imaging System (IVIS) (Perkin Elmer). Images were captured with the following settings: binning of 4 (medium), Field of View (FOV) of 24, Relative aperture at f1.2 and exposure time of 180s. Bioluminescence was calculated using Living Image software V4.1. The number of mice per dilution developing BU up to six-months was recorded. Mice were observed up to six-months. The data were plotted and a standard curve was fitted using least squares non-linear regression. The ID50 and 95% confidence intervals were interpolated with reference to the standard curve using GraphPad Prism v 7.0a.

Real time quantitative PCR

For each mosquito that blood-fed DNA was individually extracted from the dissected head, abdomen and legs of each insect using the MoBio Powersoil DNA extraction kit following manufacturer's instructions (MoBio Laboratories Inc., Carlsbad CA USA). The insects were held by the wings and each body portion (legs, abdomen, head) individually and separately removed with sterile, fine forceps to avoid cross contamination during dissections. Body parts were stored individually (legs were pooled per insect) in sterile 1.5mL tubes at -20°C until DNA extraction. DNA was similarly extracted from mouse tissue. Procedural extraction control blanks (sterile water) were included at a frequency of 10% to monitor potential PCR contamination, in addition to no-template negative controls. IS2404 quantitative PCR (qPCR) was performed as described



Fig 1. Schematic representations of the two BU transmission models tested in this study. (A) Model-1 tests transmission of *M. ulcerans* present on a skin surface following a puncturing injury created by mosquito blood-feeding or needle stick. (B) Visualization of bioluminescent *M. ulcerans* JKD8049 (harbouring plasmid pMV306 *hsp:lux*G13) [29, 30] on the mouse-tail in model-1, showing the distribution of bacteria immediately after coating for two mice, versus an uncoated animal. *M. ulcerans* culture concentration used for tail coating was 8.3x10⁵ CFU/mL.

https://doi.org/10.1371/journal.pntd.0005553.g001

[33]. IS2404 cycle threshold (Ct) values were converted to genome equivalents (GE) to estimate bacterial load within a sample by reference to a standard curve ($r^2 = 0.9312$, y = [-3.000Ln(x)+-39.33]*Z, where y = Ct and x = amount of DNA [fg] and Z = the dilution factor]), calculated using dilutions of genomic DNA from *M. ulcerans* strain JKD8049, quantified using a fluorimeter (Qubit, Invitrogen) [33].

Preparation of mouse tissue for analysis

At the end of the experimental period or when a clinical end-point was reached mice were humanely killed. The region of a mouse-tail spanning a likely lesion was cut into three equal sections for histology, qPCR and CFU counts. Individual tail pieces for CFU counts were weighed and placed into sterile 2ml screw capped tube containing 0.5g of large glass beads and 600µl of sterile 1x PBS. Tissues were homogenized using four rounds of 2x 30second pulses in a high-speed tissue-disruptor at 6500 rpm, with tubes placed on ice for 5 minutes between each round. A 300µl volume of this homogenate was decontaminated with 300µl of 2% NaOH (v/v) and incubated at room temperature for 15 minutes. The preparation was neutralized drop-wise with a 10% solution of orthophosphoric acid (v/v) with added bromophenol blue until the solution changed from blue to clear. The mixtures were diluted in PBS and CFUs determined by spot plating as described above.

Histology

Sections of mouse-tails were fixed in 10% (w/v) neutral-buffered-formalin and imbedded in paraffin. Each mouse-tail was sectioned transversely (4μ m thick) and subjected to Ziehl-Neelson and hematoxylin/eosin staining. The fixed and stained tissue sections were examined by light microscopy.

Results

M. ulcerans is efficiently transmitted to a mammalian host by mechanical means

In experiment one, we established a murine model of *M. ulcerans* transmission that represented a skin surface contaminated with the bacteria and then subjected to a minor penetrating trauma, via either a mosquito bite or needle stick puncture. For this experiment, only Aedes notoscriptus mosquitoes were used, a local species previously associated with BU in south east Australia [22]. We first coated the tails of 12 mice with M. ulcerans then exposed only the tails to A. notoscriptus. Six of the 12 mice exposed to mosquitoes were bitten once each, and of these six mice, two developed BU lesions (Table 1, Fig 1B, Fig 2A). Histology of these lesions confirmed a subcutaneous focus of AFB, within a zone of necrotic tissue. There was also characteristic epithelial hyperplasia adjacent to the site of infection (Fig 2B and 2C). Material extracted from the lesions was IS2404 qPCR-positive and culture positive for M. ulcerans (S1 Table). Mice bitten by mosquitoes but with tails coated only with sterile culture media did not develop lesions (Table 1). In the same experiment, we also subjected five mice to a single needle stick puncture. Each mouse had their tail coated with M. ulcerans as for the mosquito biting. Four of these five mice developed *M. ulcerans* positive lesions (Table 1, Fig 2D), with subcutaneous foci of infection and viable bacteria (Fig 2F). The histology of these lesions was the same as the mice subjected to mosquito blood feeding, however bacterial burden was higher following needle stick puncture (Fig 2C compared with Fig 2F). Six mice with their tails coated with M. ulcerans but not subjected to a puncturing injury did not develop lesions and remained healthy until the completion of the experiment at six months. This experiment





https://doi.org/10.1371/journal.pntd.0005553.g002

suggested that minor penetrating skin trauma (defined here as a puncture <0.5mm diameter and <2mm deep) to a skin surface contaminated with *M. ulcerans* is sufficient to cause infection. It also revealed a means by which mosquitoes could act as mechanical vectors of *M. ulcerans*.

In experiment two, using approximately the same dose of bacteria to coat the mouse-tails, we repeated experiment-1 but with *Aedes aegypti*, because of the close association of this mosquito to humans world-wide and their potential to vector pathogens. Despite multiple mosquito bites per mouse in the second experiment compared to the first, none of the five *Aedes aegypti*-exposed mice developed lesions (Table 1). However, as for experiment 1, four of five mice subjected to single, needle stick puncture developed *M. ulcerans* positive tail lesions (Table 1). A third needle stick puncture experiment was then conducted, this time using a smaller diameter, 30-gauge needle, to assess the impact of a smaller injury. There were 13/14

mice that developed BU when subjected to a single needle stick puncture through a contaminated skin surface, while eight mice with contaminated skin but no injury did not progress to disease (Table 1). Thus, across the three experiments there were 21/24 mice (88%) with needle stick puncture that developed BU, suggesting that this is an efficient mode of disease transmission (Table 1). Either *A. notoscriptus* mosquito bite or needle stick trauma significantly increased the risk of developing BU in our mouse skin surface contamination model (Tables 2 & 3).

Higher M. ulcerans burden on mosquitoes is associated with transmission

We assessed the likely burden of *M. ulcerans* by individual IS2404 qPCR of the head, abdomen and legs for each mosquito that blood fed (Fig 3). A summary of these results is shown in Fig <u>3A</u>. We noted that the bacterial load (expressed as genome equivalents [GE]) was significantly higher in the mosquito heads associated with mice that developed lesions (p<0.05) (Fig <u>3B</u>, <u>S1</u> <u>Table</u>). These data point to a threshold, above which some mosquitoes may become competent mechanical vectors for *M. ulcerans* transmission.

Estimation of incubation period and infectious dose of transmission model-1

Based on the time until a tail lesion was first observed, and when using the highest concentration of bacteria (dose range 9–55 CFU, Table 1) we estimated a median incubation period (IP) of 12 weeks (Fig 4A). This result overlaps with the IP in humans for BU, estimated in different epidemiological studies from 4–10 weeks in Uganda during the 1960s [14] and 4–37 weeks in south east Australia [28]. We then estimated the infectious dose-50 (ID50). We used six different concentrations of *M. ulcerans* to coat the tails of mice (n = 5 mice/dilution), subjected each mouse-tail to a single needle stick puncture, and then observed the number of mice for each dilution that developed Buruli ulcer, allowing an ID₅₀ estimate of 2.6 CFU (95% CI 1.6–3.6 CFU) (Fig 4B). To our knowledge this is the first estimate of an *M. ulcerans* infectious dose and indicates that like *Mycobacterium tuberculosis* and *Mycobacterium leprae*, a small quantity of this slow growing mycobacterium is sufficient to cause disease.

Discussion

Pathogen transmission by arthropods is generally characterized by either biological transmission, such as malaria [34] or mechanical transmission, where replication or biological transformation of the pathogen within the vector is not necessary for disease spread [35, 36]. Here, we show for the first time an efficient mechanical mode of transmission of Mycobacterium ulcerans to a mammalian host that implicates both puncturing injuries and arthropods. In our study, the uninfected host is externally contaminated with M. ulcerans that is certainly plausible in many areas of the world. We propose that a micro-puncture wound of any sort whether it is by natural means e.g., a thorn, arthropod bite or artificially induced via a human-mediated puncture has the potential to inject M. ulcerans and generate an ulcer. This research was designed around established frameworks for implicating vectors in disease transmission and provides the necessary causational evidence to help resolve the 80-year mystery on how M. *ulcerans* is spread to people [15, 37]. The efficient establishment of BU we have shown here via minor penetrating trauma such as a needle puncture through a contaminated skin surface helps fulfil one of four Barnett Criteria [37]. In vector ecology, mechanical transmission is defined as a non-circulative process involving accidental transport of the pathogen [36]. That is, the pathogen, in some fashion, nonspecifically associates or contaminates the mouthparts

	Developed BU	No infection	Total
Mosquito blood feed	2*	4	6
No blood feed	0	18	18
Total	2	22	24

Table 2. Correspondence between A. notoscriptus biting and development of BU for the contaminated skin model.

* Fishers exact test: odds ratio 20.56 (95% CI 0.83–507.9), p<0.05

https://doi.org/10.1371/journal.pntd.0005553.t002

(stylet) of an arthropod vector. Insect mechanical transmission of BU implies that if *M. ulcerans* were ingested and then egested via regurgitation or salivation, the mechanism would act more like a syringe than a needle [38]. Such a mode of *M. ulcerans* disease transmission is supported by previous laboratory studies in which *Naucoris* and Belostmatid water bugs were contaminated via feeding on maggot prey that had been injected with *M. ulcerans* or fed naturally on dietary contaminated larval mosquito prey [26, 39].

Our demonstration in the current study of mechanical transmission suggests there are potentially multiple or parallel pathways of *M. ulcerans* infection [37]. Examples of bacterial diseases with multiple transmission modes include tularemia, plague and trachoma [40, 41]. Support for our mechanical transmission model also comes from the many field reports over the decades of *M. ulcerans* infection following trauma to the skin. Case reports have noted BU following a suite of penetrating injuries ranging from insect bites (ants, scorpions), snake bite, human bite, splinters, gunshot, hypodermic injections of medication and vaccinations [42– 44]. Epidemiologists in Uganda during the 1960s and 70s suggested sharp-edged grasses might introduce the bacteria [45]. However, a recent laboratory study established that abrasions of the skin in Guinea pig models and subsequent application of *M. ulcerans* was not enough to cause an ulcer, however, this same study established that a subcutaneous injection would cause an ulcer [46]. As a sequel to this study in Guinea pigs, we raised the question of how likely it was that mammalian skin could be sufficiently coated in M. ulcerans that an injury from natural or anthropogenic sources could lead to infection. Other explanations for the transmission of *M. ulcerans* include linkages with human behavior that increase direct contact with human skin and contaminated water [15]. A recent study from Cameroon recorded the persistence of *M. ulcerans* over a 24-month period in a waterhole used by villagers (including BU patients) for bathing [47]. A similar study in Ghana documented a 90% positivity rate for MU for water bodies frequented by community members for bathing and washing purposes [48]. Hence, it is reasonable to envisage a scenario where a villager's skin surface becomes contaminated after bathing in such a water body and is primed for infection if (i) the concentration of bacteria is sufficiently high, and (ii) an inoculating event occurs. Whereas, in Australia, earlier studies have shown that M. ulcerans contamination of possum feces in and around the gardens of BU patients might present a similar skin surface contamination model in this region [17, 49].

Table 3. Correspondence between needle stick puncture and development of BU for the contaminated skin model.

	Developed BU	No infection	Total
Needle stick puncture	21*	3	24
No needle stick puncture	0	18	18
Total	21	21	42

* Fishers exact test: odds ratio 227.3 (95% CI 11.0-4697), p<0.0001

https://doi.org/10.1371/journal.pntd.0005553.t003



Fig 3. Summary of *M. ulcerans* **burden on mosquitoes post-feeding on contaminated mouse-tails.** (A) Visualization of the mean number of *M. ulcerans* detected per dissected mosquito segment, as assessed by IS2404 qPCR and expressed as genome equivalents (GE). 'N' indicates the total number of mosquitoes tested. Red-shaded mosquitoes transmitted *M. ulcerans*, leading to mouse-tail lesions. Green-shaded mosquitoes blood-fed on mouse-tails but lesions did not develop. (B, C, D) Plots of the individual qPCR results for each mosquito segment, listed by experiment. Red dots correspond to qPCR bacterial load for mosquitoes that transmitted *M. ulcerans* indicates the mean bacterial load per mosquito and error bars depict standard deviation. The y-axis is GE and x-axis is experiment. The qPCR data for individual insects is contained in S1 Table.

https://doi.org/10.1371/journal.pntd.0005553.g003

Future experiments will address the possibility that insect vectors may be able to move *M*. *ulcerans* from one source and inject it into an animal or human.

Our focus on mechanical mosquito transmission arose from previous surveys in southeastern Australia where a strong association between *M. ulcerans* positive mosquitoes and human cases of BU has shown that *M. ulcerans* has not only been found on adult mosquitoes from both lab and field studies but also a biological gradient, where maximum likelihood estimates (MLE) of the proportion of *M. ulcerans*-positive mosquitoes increased as the number of cases of BU increased [22, 39, 50–53]. However, a recent study in Benin, West Africa found no evidence of *M. ulcerans* in association with adult mosquitoes [54]. The authors concluded that the mode of transmission might differ between southeastern Australia and Africa. Although, laboratory and fieldwork in West Africa suggest that aquatic insects, including mosquito larvae, play a role as reservoirs in nature for *M. ulcerans* that may be indirectly tied to transmission by serving as dispersal mechanisms [21, 26, 55]. Epidemiological studies have shown that direct contact with water is not a universal risk factor for BU [8, 11]. Prior exposure to insect bites and gardening are also independent risk factors for developing BU, while use of insect repellent is protective [11, 56].



Fig 4. *M. ulcerans* **incubation period and infectious dose**₅₀. (A) Incubation period of *M. ulcerans* based on the time between sterile-needle puncture of an *M. ulcerans* contaminated mouse-tail and first observation of a lesion. (B) Estimated *M. ulcerans* ID50 for contaminated skin surface transmission model. Dashed lines indicate 95% confidence intervals. Dotted lines depicts ID50.

https://doi.org/10.1371/journal.pntd.0005553.g004

Laboratory support to show mosquitoes can be competent vectors to spread BU is important additional evidence required to satisfy accepted vector ecology criteria for implicating insects in disease transmission [15, 37]. We found that infection was established following very minor penetrating trauma. Mosquitoes, in general, feed by insertion of a stylet, sheathed within the proboscis, beneath the skin of a host. The stylet has an approximate diameter 10 µM tapering to 1 μ M at its tip and extending 1–2 mm below the skin surface. We estimated the density of M. ulcerans on the mouse-tails surface was 100-200 CFU/mm². Thus, the number of bacteria potentially injected during mosquito feeding through this contaminated surface is likely to be low, but this is consistent with our infectious dose estimates from needle-stick punctures, indicating an ID50 of only 2.6 CFU (Fig 4B). Aedes notoscriptus mosquitoes are approximately twice as large as Aedes aegypti. Larger size may imply a longer stylet length, longer blood feeding time with a deeper penetration to a depth that may initiate M. ulcerans infection more frequently than that from Aedes aegypti mosquitoes that are smaller and do not blood feed as long. Such subtle differences in mosquito morphology and behavior may indicate why mechanical transmission in any form may be uncommon. There are strong parallels here with Mycobacterium leprae, the agent of leprosy. Like BU, the mode of transmission of the leprosy bacillus is unclear, but the infective dose is known to be very low (10 bacteria) and epidemiological evidence suggests multiple transmission pathways, including entry of the bacteria after skin trauma [57, 58]. Our infective dose estimate for *M. ulcerans* is consistent with observations that pathogens producing locally acting molecules to cause disease (e.g. the polyketide toxin mycolactone of *M. ulcerans*) have lower infective doses [59].

In summary, we have uncovered a highly efficient (88% rate) for needle stick skin punctures and a lower rate for mosquito punctures, suggesting a plausible mechanical transmission mode of *M. ulcerans* infection via anthropogenic or natural skin-puncturing microtrauma. We conclude from these experiments that reduction of exposure to insect bites, access to clean water for bathing, and prompt treatment of wounds and existing BU are concrete measures likely to interrupt BU transmission.

Supporting information

S1 Table. IS2404 mosquito qPCR results. (PDF)

Author Contributions

Conceptualization: JRW PDRJ TPS.

Data curation: JRW TPS.

Formal analysis: JRW KMM JLP RM SJP TFO WZ JKA PDRJ TPS.

Funding acquisition: JRW PDRJ TPS.

Investigation: JRW KMM JLP RM SJP BH TFO WZ JKA TPS.

Methodology: JRW KMM JLP RM SJP BH TFO JKA PDRJ TPS.

Project administration: JRW PDRJ TPS.

Resources: JRW KMM JLP RM SJP BH TFO WZ JKA PDRJ TPS.

Supervision: JRW PDRJ TPS.

Validation: JRW PDRJ TPS.

Visualization: JRW KMM JLP PDRJ TPS.

Writing - original draft: JRW TPS.

Writing - review & editing: JRW KMM JLP RM SJP BH TFO WZ JKA PDRJ TPS.

References

- Anon. Second WHO report on neglected tropical diseases: sustaining the drive to overcome the global impact of neglected tropical diseases. Geneva, Switzerland: World Health Organization; 2013. p. 11– 7.
- P. M, Tolhurst JC, Buckle G, Sissons HA. A new mycobacterial infection in man. J Pathol Bacteriol. 1948; 60(1):93–122.
- Johnson PD, Stinear T, Small PL, Pluschke G, Merritt RW, Portaels F, et al. Buruli ulcer (M. ulcerans infection): new insights, new hope for disease control. PLoS Med. 2005; 2(4):e108. PubMed Central PMCID: PMC1087202. https://doi.org/10.1371/journal.pmed.0020108 PMID: 15839744
- 4. Amofah GK, Sagoe-Moses C, Adjei-Acquah C, Frimpong EH. Epidemiology of Buruli ulcer in Amansie West district, Ghana. Trans R Soc Trop Med Hyg. 1993; 87(6):644–5. PMID: 8296362
- Asiedu K, Etuaful S. Socioeconomic implications of Buruli ulcer in Ghana: a three-year review. Am J Trop Med Hyg. 1998; 59(6):1015–22. PMID: 9886216
- Debacker M, Portaels F, Aguiar J, Steunou C, Zinsou C, Meyers W, et al. Risk factors for Buruli ulcer, Benin. Emerg Infect Dis. 2006; 12(9):1325–31. PubMed Central PMCID: PMC3294727. https://doi.org/ 10.3201/eid1209.050598 PMID: 17073079
- Jacobsen KH, Padgett JJ. Risk factors for *Mycobacterium ulcerans* infection. Int J Infect Dis. 2010; 14 (8):e677–81. https://doi.org/10.1016/j.ijid.2009.11.013 PMID: 20185351
- Landier J, Boisier P, Fotso Piam F, Noumen-Djeunga B, Sime J, Wantong FG, et al. Adequate wound care and use of bed nets as protective factors against Buruli Ulcer: results from a case control study in Cameroon. PLoS Negl Trop Dis. 2011; 5(11):e1392. PubMed Central PMCID: PMC3210760. https:// doi.org/10.1371/journal.pntd.0001392 PMID: 22087346
- Nackers F, Johnson RC, Glynn JR, Zinsou C, Tonglet R, Portaels F. Environmental and health-related risk factors for *Mycobacterium ulcerans* disease (Buruli ulcer) in Benin. Am J Trop Med Hyg. 2007; 77 (5):834–6. PMID: <u>17984337</u>
- Pouillot R, Matias G, Wondje CM, Portaels F, Valin N, Ngos F, et al. Risk factors for buruli ulcer: a case control study in Cameroon. PLoS Negl Trop Dis. 2007; 1(3):e101. PubMed Central PMCID: PMC2154388. https://doi.org/10.1371/journal.pntd.0000101 PMID: 18160977
- 11. Quek TY, Athan E, Henry MJ, Pasco JA, Redden-Hoare J, Hughes A, et al. Risk factors for *Mycobacte-rium ulcerans* infection, southeastern Australia. Emerg Infect Dis. 2007; 13(11):1661–6. PubMed Central PMCID: PMC3375781. https://doi.org/10.3201/eid1311.061206 PMID: 18217548
- Raghunathan PL, Whitney EA, Asamoa K, Stienstra Y, Taylor TH Jr., Amofah GK, et al. Risk factors for Buruli ulcer disease (*Mycobacterium ulcerans* Infection): results from a case-control study in Ghana. Clin Infect Dis. 2005; 40(10):1445–53. https://doi.org/10.1086/429623 PMID: 15844067
- Sopoh GE, Barogui YT, Johnson RC, Dossou AD, Makoutode M, Anagonou SY, et al. Family relationship, water contact and occurrence of Buruli ulcer in Benin. PLoS Negl Trop Dis. 2010; 4(7):e746. PubMed Central PMCID: PMC2903473. https://doi.org/10.1371/journal.pntd.0000746 PMID: 20644620
- 14. Uganda-Buruli-Group. Epidemiology of *Mycobacterium ulcerans* infection (Buruli ulcer) at Kinyara, Uganda. Trans R Soc Trop Med Hyg. 1971; 65(6):763–75. PMID: 5157438
- Merritt RW, Walker ED, Small PL, Wallace JR, Johnson PD, Benbow ME, et al. Ecology and transmission of Buruli ulcer disease: a systematic review. PLoS Negl Trop Dis. 2010; 4(12):e911. PubMed Central PMCID: PMC3001905. https://doi.org/10.1371/journal.pntd.0000911 PMID: 21179505
- Portaels F, Meyers WM, Ablordey A, Castro AG, Chemlal K, de Rijk P, et al. First cultivation and characterization of *Mycobacterium ulcerans* from the environment. PLoS Negl Trop Dis. 2008; 2(3): e178. PubMed Central PMCID: PMC2268003. https://doi.org/10.1371/journal.pntd.0000178 PMID: 18365032
- Fyfe JA, Lavender CJ, Handasyde KA, Legione AR, O'Brien CR, Stinear TP, et al. A major role for mammals in the ecology of Mycobacterium ulcerans. PLoS Negl Trop Dis. 2010; 4(8):e791. Epub 2010/08/ 14. PubMed Central PMCID: PMC2919402. <u>https://doi.org/10.1371/journal.pntd.0000791</u> PMID: 20706592
- 18. Williamson HR, Benbow ME, Nguyen KD, Beachboard DC, Kimbirauskas RK, McIntosh MD, et al. Distribution of *Mycobacterium ulcerans* in buruli ulcer endemic and non-endemic aquatic sites in Ghana.

PLoS Negl Trop Dis. 2008; 2(3):e205. PubMed Central PMCID: PMC2268743. https://doi.org/10.1371/ journal.pntd.0000205 PMID: 18365034

- Marsollier L, Stinear T, Aubry J, Saint Andre JP, Robert R, Legras P, et al. Aquatic plants stimulate the growth of and biofilm formation by Mycobacterium ulcerans in axenic culture and harbor these bacteria in the environment. Appl Environ Microbiol. 2004; 70(2):1097–103. Epub 2004/02/10. PubMed Central PMCID: PMC348869. https://doi.org/10.1128/AEM.70.2.1097-1103.2004 PMID: 14766593
- Marsollier L, Severin T, Aubry J, Merritt RW, Saint Andre JP, Legras P, et al. Aquatic snails, passive hosts of *Mycobacterium ulcerans*. Appl Environ Microbiol. 2004; 70(10):6296–8. PubMed Central PMCID: PMC522119. https://doi.org/10.1128/AEM.70.10.6296-6298.2004 PMID: 15466578
- Portaels F, Elsen P, Guimaraes-Peres A, Fonteyne PA, Meyers WM. Insects in the transmission of *Mycobacterium ulcerans* infection. Lancet. 1999; 353(9157):986. <u>https://doi.org/10.1016/S0140-6736</u> (98)05177-0 PMID: 10459918
- Johnson PD, Azuolas J, Lavender CJ, Wishart E, Stinear TP, Hayman JA, et al. Mycobacterium ulcerans in mosquitoes captured during outbreak of Buruli ulcer, southeastern Australia. Emerg Infect Dis. 2007; 13(11):1653–60. Epub 2008/01/26. PubMed Central PMCID: PMC3375796. <u>https://doi.org/10. 3201/eid1311.061369</u> PMID: 18217547
- Lavender CJ, Fyfe JA, Azuolas J, Brown K, Evans RN, Ray LR, et al. Risk of Buruli ulcer and detection of *Mycobacterium ulcerans* in mosquitoes in southeastern Australia. PLoS Negl Trop Dis. 2011; 5(9): e1305. PubMed Central PMCID: PMC3176747. https://doi.org/10.1371/journal.pntd.0001305 PMID: 21949891
- Marsollier L, Andre JP, Frigui W, Reysset G, Milon G, Carbonnelle B, et al. Early trafficking events of Mycobacterium ulcerans within Naucoris cimicoides. Cell Microbiol. 2007; 9(2):347–55. https://doi.org/ 10.1111/j.1462-5822.2006.00790.x PMID: 16939536
- Marsollier L, Aubry J, Coutanceau E, Andre JP, Small PL, Milon G, et al. Colonization of the salivary glands of Naucoris cimicoides by *Mycobacterium ulcerans* requires host plasmatocytes and a macrolide toxin, mycolactone. Cell Microbiol. 2005; 7(7):935–43. <u>https://doi.org/10.1111/j.1462-5822.2005</u>. 00521.x PMID: 15953026
- Marsollier L, Robert R, Aubry J, Saint Andre JP, Kouakou H, Legras P, et al. Aquatic insects as a vector for *Mycobacterium ulcerans*. Appl Environ Microbiol. 2002; 68(9):4623–8. PubMed Central PMCID: PMC124085. https://doi.org/10.1128/AEM.68.9.4623-4628.2002 PMID: 12200321
- Gordon CL, Buntine JA, Hayman JA, Lavender CJ, Fyfe JA, Hosking P, et al. All-oral antibiotic treatment for buruli ulcer: a report of four patients. PLoS Negl Trop Dis. 2010; 4(11):e770. PubMed Central PMCID: PMC2994921. https://doi.org/10.1371/journal.pntd.0000770 PMID: 21152060
- Trubiano JA, Lavender CJ, Fyfe JA, Bittmann S, Johnson PD. The incubation period of Buruli ulcer (*Mycobacterium ulcerans* infection). PLoS Negl Trop Dis. 2013; 7(10):e2463. PubMed Central PMCID: PMC3789762. https://doi.org/10.1371/journal.pntd.0002463 PMID: 24098820
- Andreu N, Zelmer A, Fletcher T, Elkington PT, Ward TH, Ripoll J, et al. Optimisation of bioluminescent reporters for use with mycobacteria. PLoS One. 2010; 5(5):e10777. PubMed Central PMCID: PMCPMC2875389. https://doi.org/10.1371/journal.pone.0010777 PMID: 20520722
- Omansen TF, Porter JL, Johnson PD, van der Werf TS, Stienstra Y, Stinear TP. In-vitro activity of avermectins against *Mycobacterium ulcerans*. PLoS Negl Trop Dis. 2015; 9(3):e0003549. PubMed Central PMCID: PMCPMC4351077. https://doi.org/10.1371/journal.pntd.0003549 PMID: 25742173
- Hong H, Gates PJ, Staunton J, Stinear T, Cole ST, Leadlay PF, et al. Identification using LC-MSn of cometabolites in the biosynthesis of the polyketide toxin mycolactone by a clinical isolate of Mycobacterium ulcerans. Chem Commun (Camb). 2003;(22):2822–3. Epub 2003/12/04.
- Axford JK, Ross PA, Yeap HL, Callahan AG, Hoffmann AA. Fitness of wAlbB Wolbachia Infection in Aedes aegypti: Parameter Estimates in an Outcrossed Background and Potential for Population Invasion. Am J Trop Med Hyg. 2015.
- 33. Fyfe JA, Lavender CJ, Johnson PD, Globan M, Sievers A, Azuolas J, et al. Development and application of two multiplex real-time PCR assays for the detection of Mycobacterium ulcerans in clinical and environmental samples. Appl Environ Microbiol. 2007; 73(15):4733–40. Epub 2007/05/29. PubMed Central PMCID: PMC1951036. https://doi.org/10.1128/AEM.02971-06 PMID: 17526786
- Ross R. On some Peculiar Pigmented Cells Found in Two Mosquitos Fed on Malarial Blood. Br Med J. 1897; 2(1929):1786–8. PubMed Central PMCID: PMCPMC2408186. PMID: 20757493
- **35.** Foil LD, Gorham JR. Mechanical Transmission of Disease Agents by Arthropods. In: Eldridge BF, Edman JD, editors. Medical Entomology: A Textbook on Public Health and Veterinary Problems Caused by Arthropods. Dordrecht: Springer Netherlands; 2000. p. 461–514.
- 36. Drucker M, Then C. Transmission activation in non-circulative virus transmission: a general concept? Curr Opin Virol. 2015; 15:63–8. https://doi.org/10.1016/j.coviro.2015.08.006 PMID: 26318641

- **37.** Barnett HC, editor The incrimination of arthropods as vectors of disease. Proceedings of the 11th International Congress of Entomology; 1962.
- **38.** Harris KF. An ingestion-egestion hypothesis of noncirculative virus transmission. In: Mandahar L, editor. Plant Viruses. II. Boca Raton Inc., Fla: CRC Press; 1990. p. 177–204.
- Wallace JR, Gordon MC, Hartsell L, Mosi L, Benbow ME, Merritt RW, et al. Interaction of *Mycobacte-rium ulcerans* with mosquito species: implications for transmission and trophic relationships. Appl Environ Microbiol. 2010; 76(18):6215–22. PubMed Central PMCID: PMC2937476. <u>https://doi.org/10.1128/AEM.00340-10 PMID: 20675453</u>
- 40. Hugh-Jones M, Blackburn J. The ecology of Bacillus anthracis. Mol Aspects Med. 2009; 30(6):356–67. https://doi.org/10.1016/j.mam.2009.08.003 PMID: 19720074
- Petersen JM, Mead PS, Schriefer ME. Francisella tularensis: an arthropod-borne pathogen. Vet Res. 2009; 40(2):7. PubMed Central PMCID: PMCPMC2695023. https://doi.org/10.1051/vetres:2008045 PMID: 18950590
- Debacker M, Zinsou C, Aguiar J, Meyers WM, Portaels F. First case of *Mycobacterium ulcerans* disease (Buruli ulcer) following a human bite. Clin Infect Dis. 2003; 36(5):e67–8. <u>https://doi.org/10.1086/</u>367660 PMID: 12594656
- Hofer M, Hirschel B, Kirschner P, Beghetti M, Kaelin A, Siegrist CA, et al. Brief report: disseminated osteomyelitis from *Mycobacterium ulcerans* after a snakebite. N Engl J Med. 1993; 328(14):1007–9. https://doi.org/10.1056/NEJM199304083281405 PMID: 8450852
- Meyers WM, Shelly WM, Connor DH, Meyers EK. Human *Mycobacterium ulcerans* infections developing at sites of trauma to skin. Am J Trop Med Hyg. 1974; 23(5):919–23. PMID: 4451232
- Barker DJ. Epidemiology of *Mycobacterium ulcerans* infection. Trans R Soc Trop Med Hyg. 1973; 67 (1):43–50. PMID: 4798218
- 46. Williamson HR, Mosi L, Donnell R, Aqqad M, Merritt RW, Small PL. *Mycobacterium ulcerans* fails to infect through skin abrasions in a guinea pig infection model: implications for transmission. PLoS Negl Trop Dis. 2014; 8(4):e2770. PubMed Central PMCID: PMCPMC3983084. <u>https://doi.org/10.1371/journal.pntd.0002770 PMID: 24722416</u>
- Bratschi MW, Ruf MT, Andreoli A, Minyem JC, Kerber S, Wantong FG, et al. *Mycobacterium ulcerans* persistence at a village water source of Buruli ulcer patients. PLoS Negl Trop Dis. 2014; 8(3):e2756. PubMed Central PMCID: PMCPMC3967953. https://doi.org/10.1371/journal.pntd.0002756 PMID: 24675964
- 48. Narh CA, Mosi L, Quaye C, Dassi C, Konan DO, Tay SC, et al. Source tracking Mycobacterium ulcerans infections in the Ashanti region, Ghana. PLoS Negl Trop Dis. 2015; 9(1):e0003437. PubMed Central PMCID: PMC4303273. https://doi.org/10.1371/journal.pntd.0003437 PMID: 25612300
- Carson C, Lavender CJ, Handasyde KA, O'Brien CR, Hewitt N, Johnson PD, et al. Potential wildlife sentinels for monitoring the endemic spread of human buruli ulcer in South-East australia. PLoS Negl Trop Dis. 2014; 8(1):e2668. PubMed Central PMCID: PMC3907424. <u>https://doi.org/10.1371/journal.pntd.</u>0002668 PMID: 24498452
- Johnson PD, Lavender CJ. Correlation between Buruli ulcer and vector-borne notifiable diseases, Victoria, Australia. Emerg Infect Dis. 2009; 15(4):614–5. PubMed Central PMCID: PMC2671447. https:// doi.org/10.3201/eid1504.081162 PMID: 19331750
- Lavender CJ, Fyfe JA. Direct detection of *Mycobacterium ulcerans* in clinical specimens and environmental samples. Methods Mol Biol. 2013; 943:201–16. https://doi.org/10.1007/978-1-60327-353-4_13 PMID: 23104291
- Lavender CJ, Stinear TP, Johnson PD, Azuolas J, Benbow ME, Wallace JR, et al. Evaluation of VNTR typing for the identification of Mycobacterium ulcerans in environmental samples from Victoria, Australia. FEMS Microbiol Lett. 2008; 287(2):250–5. Epub 2008/08/30. https://doi.org/10.1111/j.1574-6968. 2008.01328.x PMID: 18754785
- 53. Tobias NJ, Seemann T, Pidot SJ, Porter JL, Marsollier L, Marion E, et al. Mycolactone gene expression is controlled by strong SigA-like promoters with utility in studies of Mycobacterium ulcerans and buruli ulcer. PLoS Negl Trop Dis. 2009; 3(11):e553. Epub 2009/11/26. PubMed Central PMCID: PMC2775157. https://doi.org/10.1371/journal.pntd.0000553 PMID: 19936295
- Zogo B, Djenontin A, Carolan K, Babonneau J, Guegan JF, Eyangoh S, et al. A Field Study in Benin to Investigate the Role of Mosquitoes and Other Flying Insects in the Ecology of *Mycobacterium ulcerans*. PLoS Negl Trop Dis. 2015; 9(7):e0003941. PubMed Central PMCID: PMCPMC4510061. <u>https://doi.org/10.1371/journal.pntd.0003941</u> PMID: 26196901
- Roche B, Benbow ME, Merritt R, Kimbirauskas R, McIntosh M, Small PL, et al. Identifying the Achilles' heel of multi-host pathogens: The concept of keystone "host" species illustrated by transmission. Environ Res Lett. 2013; 8(4):045009. PubMed Central PMCID: PMC3925833. https://doi.org/10.1088/1748-9326/8/4/045009 PMID: 24554969

- Quek TY, Henry MJ, Pasco JA, O'Brien DP, Johnson PD, Hughes A, et al. Mycobacterium ulcerans infection: factors influencing diagnostic delay. Med J Aust. 2007; 187(10):561–3. PMID: 18021043
- Bratschi MW, Steinmann P, Wickenden A, Gillis TP. Current knowledge on *Mycobacterium leprae* transmission: a systematic literature review. Lepr Rev. 2015; 86(2):142–55. PMID: 26502685
- Shepard CC, McRae DH. *Mycobacterium leprae* in Mice: Minimal Infectious Dose, Relationship between Staining Quality and Infectivity, and Effect of Cortisone. J Bacteriol. 1965; 89:365–72. PubMed Central PMCID: PMCPMC305516. PMID: 14255702
- Leggett HC, Cornwallis CK, West SA. Mechanisms of pathogenesis, infective dose and virulence in human parasites. PLoS Pathog. 2012; 8(2):e1002512. PubMed Central PMCID: PMCPMC3280976. https://doi.org/10.1371/journal.ppat.1002512 PMID: 22359500